TOFMAHARASHTRA SOLIS SINGLES SOLIS SOLI BANK & TREASURY RECEIPT (e-SBTR)

21637588565637

Bank/Branch: SBI / 11688-SME BACKBAY RECLAMATION BRANCH

Pmt Txn id: 120096

Stationery No XXXXXXXXXX5637

Pmt DtTime: 08-08-2025@05:02:23 Print DtTime: 08-08-2025@05:04:01

District: 7101/MUMBAI

Office Name: IGR182-BOM1_MUMBAI CITY 1 SUB

REGISTRAR

ChallanIdNo: 00211688080825115201 GRAS GRN: MH006860227202526S

GRN Date: 08-08-2025@05:02:24

StDuty Schm: 0030045501

StDuty Amt: Rs 1199967/-(Rs One One, Nine Nine, Nine Six Seven Only)

RgnFee Schm

RgnFee Amt:

Article:

5(h)(A)(iv)/Agreement creating right and having monetary value

Prop Mvblty: Not Applicable

Consideration Rs 599983200/-

Prop Descr:

9 LALWANI INDUSTRIAL ESTATE, 14 KATRAK ROAD, WADALA WEST, 400031

Duty Payer: PAN-AABCS1741G, SESHAASAI TECHNOLOGIES LIMITED

Other Party: PAN-AABCT3518Q, TATA AIG GENERAL INSURANCE CO LTD

Bank official-1 Name & Signature

Bank official-2 Name & Signature





महाराष्ट्र MAHARASHTRA

विल्ह्य कोषागार कार्यालय, ठाण

है JNL 2025

2025

E 7 AUG 2025

DW 662806

This Stamp paper forms an integral part of the Share Subscription Agreement dated August 8, 2025 entered between Seshaasai Technologies Limited (hereinafter referred to as the "Company") and TATA AIG General Insurance Company Limited ((hereinafter referred to as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns)



TAY.

1 Vangle





MAHARASHTRA कोषामार कार्यालय ठाणे कार कार किया । लिपीक 3

2025

52AB 845887

This Stamp paper forms an integral part of the Share Subscription Agreement dated August 8, 2025 entered between Seshaasai Technologies Limited (hereinafter referred to as the "Company") and TATA AIG General Insurance Company Limited ((hereinafter referred to as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns)



SEE BY





MAHARASHTRA विन्य को गगार कार्यालय, ठाण 2025

1000

AND ST

1300

TEE

XXX

5220

XX

2025

DW 662805

This Stamp paper forms an integral part of the Share Subscription Agreement dated August 8, 2025 entered between Seshaasai Technologies Limited (hereinafter referred to as the "Company") and TATA AIG General Insurance Company Limited ((hereinafter referred to as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns)





महाराष्ट्र MAHARASHTRA

विभागार

विभाग

① 2025 **①**

E 7 AUG 2025

52AB 845888

This Stamp paper forms an integral part of the Share Subscription Agreement dated August 8, 2025 entered between Seshaasai Technologies Limited (hereinafter referred to as the "Company") and TATA AIG General Insurance Company Limited ((hereinafter referred to as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns)



CONT.

THEFT

(den)

STATE OF

XXXX

CENT.

MARE.

1

k. Vary Lie

Semologist Control of the Control of

SHARE SUBSCRIPTION AGREEMENT

THIS AGREEMENT is made on August 08, 2025 ("Execution Date"), by and between:

1. ENTITIES DESCRIBED UNDER SCHEDULE 1, (hereinafter referred to as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns) of the FIRST PART;

AND

2. Seshaasai Technologies Limited, a company incorporated under the Laws of India under the provisions of the Companies Act, 1956, having company identification number U21017MH1993PLC074023 and having its registered office at 9, Lalwani Industrial Estate 14, Katrak Road, Wadala (West), Mumbai, Maharashtra, India, 400031, hereinafter referred to as the "Company" (which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns) of the SECOND PART.

The Subscriber and the Company shall hereinafter, collectively, be referred to as "Parties" and individually, as a "Party".

WHEREAS:

- A. The Company is a technology driven multi-location solutions provider offering payments solutions, communications and fulfilment solutions and IoT solutions and is proposing to launch an initial public offer ("**IPO**") of its securities. The Company has filed a draft red herring prospectus with the Securities Exchange Board of India ("**SEBI**") in relation to such IPO on December 27, 2024 ("**DRHP**").
- B. As of the Execution Date, the authorized share capital of the Company is ₹ 1,62,50,00,000 divided into 16,25,00,000 Equity Shares of ₹ 10 each and the paid-up share capital of the Company is ₹ 1,47,61,65,000 divided into 14,76,16,500 Equity Shares of ₹ 10 each. The shareholding pattern of the Company as at the Execution Date is set out in Part A of Schedule 4 (Shareholding Pattern of the Company As At The Execution Date).
- C. The Company, as a part of its pre-IPO placement, is offering its Equity Shares for subscription by the Subscriber and the Subscriber are desirous of making a financial investment in the Company by subscribing to the Subscription Shares for the Price, as set out against their respective names in **Schedule 2** (Subscription Shares Details).
- D. The Subscription Shares are proposed to be listed on the stock exchange(s) as part of IPO of the Company.
- E. The Parties are now entering into this Agreement to define their mutual rights and obligations in relation to the issue and allotment of the Subscription Shares to the Subscriber by the Company and other matters in connection therewith.

NOW, THEREFORE, THE PARTIES AGREE AS FOLLOWS:

1. DEFINITION AND INTERPRETATION

Unless a contrary intention appears and/or the context otherwise requires, in addition to the terms defined elsewhere in this Agreement, the definitions listed in **Schedule 3** (Definitions and

MUMBAI DAN PARTIE TO THE PARTI

THE CHNOLOGIES

Interpretations) shall apply throughout this Agreement. The interpretation and/or construction of this Agreement shall be in accordance with the rules of interpretation set out in Schedule 3 (Definitions and Interpretation).

2. SUBSCRIPTION SHARES

2.1 The Subscriber have agreed to subscribe to the Subscription Shares, as set out in Schedule 2 (Subscription Shares Details) in accordance with this Agreement, basis the representations and warranties provided by the Company, and subject to receipt of the offer letter in Form PAS-4 in the format as set out in the Companies (Prospectus and Allotment of Securities) Rules, 2014 ("Form PAS-4"), from the Company and completion of the other actions as specified in Clause 4 (Conditions Precedent) below.

3. PRICE

- The Subscriber has agreed to pay the Price, in the proportion as set out in Schedule 2 (Subscription 3.1 Shares Details), to collectively subscribe to the Subscription Shares, for a total aggregate subscription amount of ₹ 59,99,83,200 (Rupees Fifty Nine Crores Ninety Nine Lakhs Eighty Three Thousand Two Hundred only). The resultant shareholding of the Subscriber in the Company, post completion of investment of the Price into the Company by the Subscriber shall be as set out in Part-B of Schedule 4 (Shareholding Pattern of the Company as on the closing date).
- The issuance of the Subscription Shares shall be made by the Company on a private placement basis, 3.2 in accordance with applicable Law.
- The Subscriber shall pay, by electronic funds transfer, the Price on the Closing Date to the following 3.3 bank account of the Company ("Designated Bank Account"), for the issuance and allotment of the Subscription Shares to the Subscriber on the Closing Date:

Account No:

000405164258

Name:

Seshaasai Technologies Limited-Monitoring Account

Customer ID:

602657624

IFSC code:

ICIC0000004

Branch:

ICICI Bank Limited, 215, Free Press House,

Free Press Marg, Nariman Point, Mumbai, 400021

Bank Name:

ICICI Bank Limited

Immediately upon receipt of the Price, the Company shall submit to the Subscriber a written 3.4 confirmation of receipt showing the date and amount of the Price received.

4. CONDITIONS PRECEDENT

The obligation of the Subscriber to subscribe to the Subscription Shares shall be conditional upon 4.1 each of the following conditions (each a "Condition Precedent") having been fulfilled by the Company, to the reasonable satisfaction of the Subscriber, before the Closing Date:



2

- the Company having passed necessary resolutions of the Board and shareholders (via a special resolution), as specified under the Act and the Articles of Association of the Company for: (i) approving the issue and allotment of the Subscription Shares to the Subscriber for the Price as contemplated in this Agreement; and (ii) approving the draft letter of offer in Form PAS 4;
- (b) delivery of certified true copies of the aforesaid resolutions of the Board and shareholders of the Company to the Subscriber;
- (c) the Company having issued an offer or invitation to the Subscriber to subscribe to the Subscription Shares through issue of a private placement offer letter in the format as set out in Form PAS-4;
- (d) The Company having filed, prior to the issuance of the offer letter in Form PAS-4, Form MGT-14 as prescribed under the Companies (Management and Administration) Rules, 2014 with the jurisdictional registrar of companies in respect of approving issuance of Equity Shares to the Subscriber; and
- (e) together with Form PAS-4, Company to deliver to the Subscriber, a valuation certificate issued by Registered Valuer, certifying the fair value of the Subscription Shares determined and issued in accordance with, and as required under the Act (and rules thereunder) and on which the Subscriber can rely.
- 4.2 If any of the Conditions Precedent are not satisfied, the Subscriber may, in writing, amend or extend the timeline for completion of any of the Conditions Precedent, by notice in writing to the Company, except to the extent of any mandatory requirements under applicable Law.
- 4.3 If at any time the Company becomes aware of any circumstances that will or are likely to give rise to the non-fulfilment of the Conditions Precedent by the Closing Date the Company shall inform the Subscriber in writing. In such case, the Parties shall co-operate fully with a view to procuring alternate arrangements to give effect to the commercial understanding of the Parties in relation to the transactions contemplated under this Agreement.
- 4.4 Upon completion of the Conditions Precedent as aforesaid, the Company shall issue a certificate in the form annexed as **Schedule 5** ("**CP Satisfaction Certificate**") enclosing, where applicable, documentary evidence including certified true copies of all necessary documents evidencing fulfilment of each of the Conditions Precedent.
- Subject to receipt of the CP Satisfaction Certificate and the accompanying documents evidencing fulfilment of all of the Conditions Precedent in accordance with the terms hereof to the reasonable satisfaction of the Subscriber, unless deferred or prescribed as condition subsequent to Closing, in each case in writing, at the Subscriber' discretion, the Subscriber shall, within 2 (two) Business Days of the receipt of the CP Satisfaction Certificate, provide to the Company a written confirmation of completion of the Conditions Precedent in the form annexed as **Schedule 6** ("CP Confirmation Certificate").
- 4.6 Upon fulfilment of all the Conditions Precedent, including the other obligations under this Clause 4 (*Conditions Precedent*) the Parties shall immediately proceed to Closing.

5. CLOSING

Subject to fulfilment of the Conditions Precedent (or deferral or prescription as condition subsequent to Closing, in each case in writing, by the Subscriber) in accordance with Clause 4 (*Conditions Precedent*), the Closing shall take place at the registered office of the Company, or such other place as may be mutually agreed between the Parties. The "Closing Date" shall be the date on which Closing occurs, which date shall be: (a) a date that is not later than three (3) Business Days from the Execution

WUMBAI WOOM A THE TOTAL HAS BEEN COMMON TO TH

WANDOO SEE

Date; or (b) such other date as may be mutually agreed, in writing, between the Parties, but not later than the Long Stop Date.

- 5.2 On the Closing Date, the events set out in the following provisions of this Clause 5 (*Closing*) shall take place in the sequence set out below, each one conditional upon the other and not to be treated as mutually exclusive. Further, all transactions contemplated by this Agreement to be consummated at Closing shall be deemed to have occurred simultaneously and no such action shall be deemed to be consummated unless all such actions have been consummated. Closing shall be deemed to have occurred when each of the activities listed herein below have been completed and the Closing Date shall be deemed to be the date on which the last of the actions below are completed:
 - (a) The Subscriber shall remit the Price into the Designated Bank Account of the Company, by way of a wire transfer through normal banking channels;
 - On the Closing Date, and simultaneously upon receipt of the Price, the Board or the Committee of the Board shall pass necessary resolutions, to the reasonable satisfaction of the Subscriber: (i) approving the issue and allotment of the Subscription Shares to the Subscriber, in dematerialized form, free of all Encumbrances; (ii) authorize issuance of instructions to the Company's depository participant to credit the Subscription Shares to the dematerialized account of the Subscriber (details of which are contained in **Schedule 1** (Details of Subscriber) below); and (iii) authorize the preparation and submission of the relevant filings related to the issuance and allotment of the Subscription Shares. The Company shall hand over to the Subscriber a certified true copy of the aforesaid resolution of the Board or the Committee of the Board.
 - (c) The Company shall make the necessary applications and filings with the Company's Depository Participant, providing irrevocable instructions for the issue of Subscription Shares to the Subscriber. The Company shall provide a copy of these instructions to the Subscriber.
- 5.3 If the Closing has not occurred on or prior to the Long Stop Date, the Agreement shall terminate automatically (without any further action on the part of any of the Parties), unless Parties mutually agree in writing to extend the Long Stop Date in accordance with the terms hereof.

6. POST CLOSING OBLIGATION

- The Company will take all reasonable efforts to ensure that the Subscription Shares are allotted by the Company's Depository Participant to the Subscriber no later than five (5) Business Days from the Closing Date. On allotment of Subscription Shares, the Company will provide an updated statement of beneficiary position to the Subscriber.
- Within three (3) Business Days after Closing, the Company shall complete all post-issue filings and compliances related to the issuance of the Subscription Shares within the time statutorily prescribed for the same, including filing the return of allotment with the registrar of companies in Form PAS-3.
- 6.3 The Company agrees that the funds disbursed by the Subscriber for the subscription of the Subscription Shares shall not be utilized by the Company unless the allotment of the Subscription Shares have been completed and the return of allotment is filed with the jurisdictional registrar of companies, in accordance with the applicable Law.
- The Company shall provide the Subscriber or any of their authorised representatives and advisers such assistance, documentation and information and undertake all actions as may be reasonably required in connection with the filings and disclosures required or agreed to be made to the registrar of companies, SEBI, stock exchanges and any other disclosures required to be made under applicable Law pertaining to the transactions contemplated by this Agreement.

WINSURANCE COMMENT AND THE STATE OF THE STAT

STECHNOLOGIES LAND

7. NON-COMPLETION OF INITIAL PUBLIC OFFERING

In the event the Subscribed Shares proposed to be issued are not listed on the NSE or the BSE within 60 Business Days (or such longer period as may be mutually agreed between the Parties) from the Closing Date ("Listing Period"), if requested by the Subscriber (under this Agreement) within 30 (thirty) days of closure of the Listing Period, and subject to Applicable Law, the Company shall, at their discretion, either:

- (a) purchase back the Subscribed Shares ("Buyback") and take all necessary steps, subject to Applicable Law to undertake the Buyback at the Price paid by the Subscriber in relation to the Subscribed Shares; or
- (b) facilitate a purchase of the Subscribed Shares by a third party at the Price paid by the Subscriber in relation to the Subscribed Shares.

8. COMPANY WARRANTIES

- 8.1 The Company represents and warrants to each of the Subscriber that, as at the date of this Agreement and on the Closing Date:
 - (a) it is validly incorporated, in existence and duly registered under the Laws of its jurisdiction and has full power and authority to conduct its business as conducted;
 - (b) the Company is authorised by its articles of associations and has obtained all corporate authorisations and has all governmental, statutory, regulatory or other consents, licences and authorisations required to empower it to execute, deliver and perform its obligations under this Agreement and execution, delivery and performance of this Agreement constitutes legal, valid and binding obligation of the Company, enforceable against itself in accordance with its terms;
 - execution, delivery and performance of this Agreement will not (i) conflict with or violate any provision of its memorandum and articles of association, by-laws or equivalent constitutional documents, (ii) conflict with or contravene the provisions of or constitute a default under any documents, contracts, agreements, arrangements or any other instrument to which the Company is a party or which is applicable to it; (iii) result in a breach of any applicable Laws or regulations or of any order, decree or judgment of any court or any governmental or regulatory authority, applicable to the Company, or by which its assets and properties are bound or affected; or (iv) require the Company to obtain any authorisations, consents, waivers, registrations, declarations or filings with, or notice to any relevant authority or any person, except as expressly set forth in this Agreement or unless otherwise already obtained by the Company, copies of which have been delivered to the Subscriber;
 - issuance and allotment of the Subscription Shares by the Company to the Subscriber shall be duly and validly authorized by all necessary corporate actions of the Company and when issued, allotted and delivered in accordance with the terms of this Agreement, the Subscription Shares shall be duly and validly issued, fully paid and free and clear of any Encumbrance;
 - (e) there is no litigation, pending or threatened in writing, against the Company, challenging the validity or propriety of this Agreement or the transactions contemplated hereby or otherwise preventing the Company from entering into this Agreement and performing its obligations hereunder;
 - (f) it has the requisite power and authority to enter into and perform this Agreement;
 - (g) the Subscription Shares are duly issued and fully paid; and the Subscription Shares constitute

MUMBAI OIL HIS CE

ECHNOZOGE SING

0.94% of the shareholding of the Company immediately upon Closing. The shareholding percentage specified in **Part A** (Shareholding Pattern of the Company As At The Execution Date) and **Part B** (Shareholding Pattern As On The Closing Date) of **Schedule 4** provides true, correct and complete representations of the shareholding of the Company, constituting 100% (one hundred per cent) of the issued, subscribed, allotted and paid-up share capital of the Company, on a fully diluted basis, as of the Execution Date and the Closing Date, respectively;

- (h) the Subscription Shares will be issued in dematerialized form;
- (i) the Board has authorised the IPO of the Equity Shares of the Company and that the Company has filed the DRHP in relation to such IPO on December 27, 2024; and
- (j) the Company has not provided any unpublished price sensitive information to the Subscriber in contravention to the provisions of appliable Laws.
- 8.2 The Company agrees and undertakes to notify the Subscriber, in writing and promptly, if it becomes aware of any fact, matter or circumstance (whether existing as on the date of this Agreement or arising afterwards) which would cause any of the representation and warranties given by the Company under this Agreement, to become untrue, inaccurate or misleading in any material respect.

9. SUBSCRIBER WARRANTIES

- 9.1 Each Subscriber warrants to the Company that as at the date of this Agreement and on the Closing Date:
 - it is validly incorporated, in existence and duly registered under the Laws of its jurisdiction and has full power and authority to conduct its business as conducted;
 - (b) the execution, delivery and performance of this Agreement has been duly authorised and does not require any further corporate approvals and/ or authorizations and it has all necessary approvals required to be obtained by it for the execution, delivery and performance of this Agreement and has been validly executed by a duly authorised representative of the Subscriber. This Agreement constitutes a legal, valid and binding obligation of the Subscriber enforceable against it, in accordance with its terms;
 - (c) it has the requisite power and authority to enter into and perform this Agreement; and
 - execution, delivery and performance of this Agreement will not (i) conflict with or violate any provision of its memorandum and articles of association, by-laws or equivalent constitutional documents or (ii) conflict with or contravene the provisions of or constitute a default under any documents, contracts, agreements, arrangements or any other instrument to which the Subscriber is a party or which is applicable to it; (iii) result in a breach of any Laws or regulations or of any registration, license, permit, consent, order, decree or judgment of any court or any governmental or regulatory authority, applicable to the Subscriber, or by which its assets and properties are bound or affected; or (iv) require the Subscriber to obtain any authorisations, consents, waivers, registrations, declarations or filings with, or notice to any relevant authority or any person, except as expressly set forth in this Agreement or unless otherwise already obtained by the Subscriber, copies of which have been delivered to the Company;
 - (e) there is no litigation, pending or threatened in writing, against the Subscriber, challenging the validity or propriety of this Agreement or the transactions contemplated hereby or otherwise preventing the Subscriber from entering into this Agreement and performing its obligations hereunder; and

WINDAI WE COM WINDAI WE COM WHAT YOU IN HERE COM WINDAI WE COM WINDAI WE

Second Se

6

(f) it has read and taken note of the factual information set out in the DRHP filed by the Company on December 27, 2025.

10. INDEMNIFICATION

- 10.1 Subject to Clause 8 (Company Warranties) and Clause 9 (Subscriber Warranties), each Party ("Indemnifying Party") shall indemnify, defend and hold harmless the other Parties and their respective investment manager, trustees, directors, officers, representatives, employees and agents as applicable, ("Indemnified Persons") against any and all direct and actual losses, damages, charges, and reasonable costs and expenses, fines, interest and penalties, imposed on, sustained, incurred or suffered by the Indemnified Persons on or after the Closing Date, directly arising out of, or directly in connection with:
 - (a) any breach of, or misrepresentation in, any of the representations and warranties of the Company as set out in Clause 8.1 (*Company Warranties*) or Subscriber as set out in Clause 9.1 (*Subscriber Warranties*), as applicable;
 - (b) covenants and undertaking provided by the Indemnifying Party under this Agreement; and/or
 - (c) wilful misconduct, fraud, or gross negligence by the Indemnifying Party in connection with the transactions contemplated under this Agreement.

Each, an "Indemnity Event".

- To the extent the payment by the Indemnifying Party of any indemnification payment pursuant to the provisions of Clause 10 (*Indemnification*) is subject to receipt of approvals by the Indemnifying Party from governmental authorities, the Indemnifying Party shall apply to such governmental authorities to procure all such approvals and shall make all applications and take all steps required to obtain the same and the Indemnified Parties shall provide all necessary documents, support and co-operation as requested by the Indemnifying Party.
- 10.3 Notwithstanding anything contained in this Agreement, the Indemnifying Party shall not be liable to indemnify an Indemnified Party for any amount in excess of the portion of the Price paid by the respective Indemnified Party, save and except if the Indemnity Event is as set out in Clause 10.1 (c) (Indemnification).
- Notwithstanding anything contained in this Agreement, each Indemnified Person expressly agrees and acknowledges that the indemnification rights contained in this Clause 10 (*Indemnification*) are conditional upon the said indemnification rights being the sole monetary remedy for each Indemnified Person and accordingly, each Indemnified Person hereby expressly and specifically waives any right that it or any of them may have under Law, contract or equity to any other form of monetary remedy. Subject to occurrence of Closing, the indemnity under Clause 10.1(a) and 10.1(c) shall survive for a period of 10 years from Closing Date.
- 10.5 Upon the occurrence of an indemnifiable event as set out in Clause 10.1 (*Indemnification*), where the Indemnified Party is a Subscriber, the amount payable shall be grossed up to reflect the percentage of shares held by such Subscriber on a fully diluted basis at the time of such claim by the Subscriber.
- 10.6 <u>General procedure for claims</u>. If any Indemnified Person seeks indemnification under this Agreement, it shall, within a period of 30 (thirty) days from the occurrence of the Indemnity Event, notify in writing (the "Claim Notice") the Indemnifying Person of the Losses for which the Indemnified Person is asserting an indemnification claim ("Indemnification Claim Amount"). The Parties agree that a delay in providing the Claim Notice under this Clause 10.6 will not relieve the Indemnifying Person of its indemnification obligation, save and except to the extent of any increase in Loss caused due to

THE CHNOLOGIES IN

the failure of the Indemnified Person to provide the Claim Notice as per the timelines prescribed in this Clause 10.6 Provided that in the event such delay by the Indemnified Person has been the sole cause of the complete failure of to defend the Claim, then the Loss arising out of such Claim shall not be required to be indemnified by the Indemnifying Person.

A claim set forth in a Claim Notice will be conclusively deemed to be the liability of the Indemnifying Person (and payable by the Indemnifying Person) if: (a) the Indemnifying Person does not deliver the Claim Dispute Notice; or (b) the liability of the Indemnifying Person in respect of such claim is resolved by mutual agreement of the Indemnifying Person and the Indemnified Person. The Indemnifying Person shall pay the Indemnification Claim Amount to the Indemnified Person within 30 (thirty) days after such claim becomes payable in accordance with this Clause 10.7.

10.8 Procedure for Third Party Claims.

- After receipt by the Indemnified Person of notice of the commencement of any action, suit, (a) arbitration or other proceeding in relation to a Loss or an Indemnity Event involving a third party including a proceeding before a Governmental Authority in connection with an Indemnity Event (a "Third Party Claim"), such Indemnified Person shall as soon as commercially practicable, but in any event within 15 (fifteen) days of receipt of such Third Party Claim, give written notice of such Third Party Claim ("Third Party Claim Notice") to the Indemnifying Person providing all details, to the extent available, regarding the circumstances then known to the Indemnified Person. The Parties agree that a delay in providing the Third Party Claim Notice under this Clause 10.8(a) will not relieve the Indemnifying Person of its indemnification obligation in respect of a Third Party Claim, save and except to the extent of any increase in Loss caused due to the failure of the Indemnified Person to provide the Third Party Claim Notice as per the timelines prescribed in this Clause 10.8(a). Provided that in the event such delay by the Indemnified Party has been the sole cause of the complete failure of the defense of the Third Party Claim, then the Loss arising out of such Third Party Claim shall not be required to be indemnified by the Indemnifying Person.
- (b) The Indemnifying Person shall, subject to Clause 10.8(d), have the right, exercisable by issuing a written notice to the Indemnified Person promptly, and in any event within 15 (fifteen) Business Days from the date of receipt of the Third Party Claim Notice, to assume the sole defence of any Third Party Claim at such Indemnifying Person's expense, provided that the Indemnifying Person shall: (i) consult, in good faith, with the Indemnified Person in relation to the conduct of the Third Party Claim; (ii) keep the Indemnified Person appropriately informed of all matters pertaining to such action
- Where the Indemnifying Person elects to assume the defence of the Third Party Claim, the Indemnified Person shall provide reasonable cooperation, at the Indemnifying Person's expense, with the Indemnifying Person and its legal advisors/ representatives. Where the Indemnifying Person is assuming defence of the Third Party Claim, the Indemnified Persons shall exercise reasonable efforts to provide all the relevant information and documents, and the representatives of the Indemnified Persons be reasonably available, for the Indemnifying Persons, its counsel and advisors for conducting the defence of the Third Party Claim, within a reasonable period of time, bearing in mind the overall timelines and strategy with respect to the defence of the Third Party Claim. The Indemnifying Person will not enter into any settlement, compromise, order or judgement in respect of any such Third Party Claim without the prior written consent of the Indemnified Person, which consent shall not be unreasonably withheld.
- (d) Indemnifying Person will not have the right under Clause 10.8(b) to assume the sole defence of any Third Party Claim where: (i) the Third Party Claim seeks an injunction or equitable relief against any of the Indemnified Persons which restricts them from conducting their activities in the ordinary course of business; or (ii) the Third Party Claim involves a criminal liability on any of the Indemnified Persons; or (iii) the Indemnifying Person is not entitled to take up the

MUMBAI WILL & OIL IN

2 CHNO OS FESSO OS SESSO OS SE

defense under Applicable Law.

- (e) If the Indemnifying Person elects not to assume defence of the Third Party Claim, or the Indemnifying Person is not entitled to assume defence of the Third Party Claim as per Clause 10.8(d), the Indemnified Person will have the right to defend such Third Party Claim at the expense of the Indemnifying Person, and the Indemnifying Person shall cooperate, at its own expense, with the Indemnified Person and its legal advisors, in relation thereto. The Indemnified Person shall keep the Indemnifying Person appropriately informed of all matters pertaining to such Third Party Claim and shall conduct such Third Party Claim in a reasonable manner. The Indemnifying Person shall be deemed to have consented to and be bound by the manner in which such defence is assumed by Indemnified Person and the same will be binding upon the Indemnifying Person, and the Indemnified Person may enter into any judgment, compromise or settlement without the prior written permission of the Indemnifying Person (without prejudice to the indemnification rights of the Indemnified Person under this Agreement).
- (f) The Parties agree to cooperate with each other in connection with the status, defence, negotiation and/or settlement of any such Third Party Claim. Where the Indemnified Person is assuming sole defence of the Third Party Claim, the Indemnifying Persons shall ensure that all the relevant information and documents be provided, and the representatives of the Company/ Seller (as applicable) be reasonably available, for the Indemnified Persons, its counsel and advisors for conducting the defence of the Third Party Claim, within a reasonable period of time, bearing in mind the overall timelines and strategy with respect to the defence of the Third Party Claim.
- (g) It is hereby clarified that the Indemnifying Persons' obligation to indemnify the Indemnified Person in relation to a Third Party Claim pursuant to this Agreement will arise within 30 (thirty) days of the Indemnified Person incurring any actual expense in relation to such Third Party Claim (including costs and expenses for defending any such Third Party Claim (which defence will be undertaken as per this Agreement) and/or payments made pursuant to an order (interim or final), judgment, settlement or compromise (subject to such settlement or compromise being in accordance with this Agreement)).
- 10.9 Recovery. Notwithstanding anything contained herein, in the event that any payment/deposit needs to be made to any Person (including any Governmental Authority) in relation to any indemnity claim pursuant to applicable Law, order and/or judgement, then the Indemnifying Person shall ensure that it makes all such payments as instructed by the Indemnified Person prior to the relevant due date to enable the Indemnified Person to make such payment on the demand due date.
- 10.10 <u>Miscellaneous</u>. Notwithstanding anything contained in this Agreement:
 - (a) The Indemnifying Person shall be required to indemnify the Indemnified Person only for actual Losses incurred or suffered. However, the Indemnifying Persons will not be liable for any loss of profit or loss of opportunity incurred/ suffered by the Indemnified Persons.
 - (b) The Indemnified Persons shall not be entitled to recover more than once in respect of the same Loss.
 - (c) The Indemnifying Persons shall not be liable to indemnify the Indemnified Persons under this Agreement for a Loss if the failure or breach giving rise to such Loss is capable of remedy and the same has been remedied within a period of 30 (thirty) days from the date of notice provided by the Indemnified Persons to the Indemnifying Persons in writing of any claim for indemnity.

11. CONFIDENTIALITY

11.1 The Parties hereto acknowledge that the terms of this Agreement, including its existence shall be

9

considered confidential information and shall not be disclosed by the Parties hereto to any third party except (i) any disclosure in relation to the IPO, including disclosures to be made in any Offer Document filed with SEBI, ROC or any other governmental or regulatory authority; (ii) any disclosure of information required by applicable Law, or any governmental or regulatory authority not in relation to the IPO; (iii) to its affiliates and investors that are bound by appropriate confidentiality obligations; (iv) the disclosure of information, which, at the date of disclosure, is in the public domain; (v) any disclosure of information to a Party's professional advisors being subject to the confidentiality obligations contained in this Agreement; and (vi) information which is independently developed by the recipient from information provided or obtained other than pursuant to or in connection with this Agreement.

12. ASSIGNMENT

- 12.1 Unless the Parties agree in writing, no person shall assign, transfer, charge or otherwise deal with all or any of its rights under this Agreement nor grant, declare, create or dispose of any right or interest in it, and any purported assignment in contravention of this Clause 12 (*Assignment*) shall be void.
- Notwithstanding anything contained in this Agreement, the provisions of this Clause 12 (*Assignment*) shall fall away on listing of equity shares of the Company.

13. TRANSFER

Subject to lock-in requirements under applicable Law, the Subscription Shares are freely transferable and are free from any restrictions under the articles of association of the Company or any other agreement to which the Company is a party.

14. FURTHER ASSURANCES

- Each of the Parties shall, execute such further documents as may be required by Law or as may be necessary to implement and give effect to this Agreement.
- Each of the Parties shall comply with all obligations under this Agreement which are expressed to apply to the respective Parties.

15. COSTS

- 15.1 The Parties shall each be responsible for their own costs, charges and other expenses incurred in connection with the Proposed Transaction. The Company shall be responsible to bear the stamp duty payable on this Agreement only to the extent that the document is executed and retained in Mumbai.
- 15.2 The stamp duty payable in relation to the issuance of the Subscription Shares (including on the allotment) shall be borne by the Company.

16. NOTICES

- Any notice in connection with this Agreement shall be in writing in English and delivered by hand, e-mail, fax, registered post or courier using an internationally recognised courier company. A notice shall be effective upon receipt and shall be deemed to have been received (i) at the time of delivery, if delivered by hand, registered post or courier or (ii) at the time of transmission if delivered by e-mail or fax provided that in either case, where delivery occurs outside Working Hours, notice shall be deemed to have been received at the start of Working Hours on the next following Business Day.
- 16.2 The addresses and fax numbers of the Parties for the purpose of Clause 16.1 (*Notices*) are:

a) If to Subscriber:



ATTENNO TO CHANGE OF THE PARTY OF THE PARTY

Name: Tata AIG General Insurance Company Limited

Address: Peninsula Business Park, Tower A, 15th Floor, G.K. Marg, Lower Parel, Mumbai-400

013

Attention: Vinay Rao (Chief Investment Officer)

Email: vinay.Rao@tataaig.com

b) If to the Company:

Seshaasai Technologies Limited (formerly Seshaasai Business Forms Limited) Name:

9, Lalwani Industrial Estate 14, Katrak Road, Wadala (West), Mumbai - 400 031, Address:

Maharashtra, India

Attention: Manali Siddharth Shah

Email: companysecretary@seshaasai.com

17. TERM AND TERMINATION

This Agreement shall come into full force and effect from the Execution Date and shall remain valid 17.1 and binding on the Parties until such time that it is terminated in accordance with Clause 17.2 (Term and Termination).

This Agreement shall be terminated in the following manner: 17.2

17.2.1. Prior to Closing.

- in accordance with Clause 5.3 (Closing) i.e., if the Closing has not occurred on or prior to the (a) Long Stop Date; or
- (b) By mutual written agreement of the Parties.

17-.2.2 Post Closing.

- (a) Upon listing of the Equity Shares on the National Stock Exchange of India Limited or the BSE Limited.
- Notwithstanding anything provided in this clause or otherwise in this Agreement, the termination of 17.3 this Agreement shall not relieve any Party of any obligations or liabilities accrued prior to the date of termination.
- It is hereby agreed that in the event the Agreement is terminated prior to Closing, in terms of Clause 17.4 17.2.1 above, the Parties undertake to use their reasonable efforts to unwind all actions initiated or taken under the provisions of this Agreement in order to achieve Closing, as soon as possible.
- Clause 1 (Definitions and Interpretation), Clause 11(Confidentiality), Clause 15 (Costs), Clause 16 17.5 (Notices), Clause 17 (Term and Termination), Clause 18 (Whole Agreement and No Deed of Adherence), Clause 19 (Waivers, Rights and Remedies), Clause 20 (Counterparts), Clause 21 (Variations), Clause 22 (Severability), Clause 23 (Governing Law and Jurisdiction) shall survive termination of this Agreement.

WHOLE AGREEMENT AND NO DEED OF ADHERENCE 18.



- 18.1 This Agreement sets out the whole agreement between the Parties in respect of the subscription Shares and supersede any prior agreement (whether oral or written) relating to the transaction contemplated under this Agreement.
- The Company hereby confirms that based on the waiver letter received from the existing investors of the Company, the Subscriber need not execute a deed of adherence to the Share Subscription Agreement.

19. WAIVERS, RIGHTS AND REMEDIES

19.1 Except as expressly provided in this Agreement, no failure or delay by any party in exercising any right or remedy relating to this Agreement shall affect or operate as a waiver or variation of that right or remedy or preclude its exercise at any subsequent time. No single or partial exercise of any such right or remedy shall preclude any further exercise of it or the exercise of any other remedy.

20. COUNTERPARTS

20.1 This Agreement may be executed in any number of counterparts, and by each party on separate counterparts. Each counterpart is an original, but all counterparts shall together constitute one and the same instrument. The delivery of a signed counterparts of this Agreement by facsimile transmission or e-mail in "portable document format" (.pdf) shall be as effective as signing and delivering the counterparts in person.

21. VARIATIONS

21.1 No amendment of this Agreement shall be valid unless it is in writing and duly executed by or on behalf of all of the Parties to it.

22. SEVERABILITY

22.1 Each of the provisions of this Agreement is severable. If any such provision is held to be or becomes invalid or unenforceable in any respect under the Law of any jurisdiction, it shall have no effect in that respect and the Parties shall use all reasonable efforts to replace it in that respect with a valid and enforceable substitute provision, the effect of which is as close to its intended effect as possible.

23. GOVERNING LAW AND JURISDICTION

- 23.1 This Agreement and any obligations arising out of or in connection with this Agreement shall be governed by and interpreted in accordance with Indian Law, without regard to conflict of law principles.
- Any dispute or differences between the Parties ("Disputing Parties") arising out of or in connection with this Agreement, including any question regarding its existence, validity or termination, and any dispute relating to any obligations arising out of or in connection with it (the "Dispute") shall be referred to and finally resolved by arbitration in accordance with the provisions of the Arbitration and Conciliation Act, 1996 or any statutory modification or re-enactment thereof for the time being in force ("Rules"). If a Disputing Party gives the other Disputing Party(ies) notice that a Dispute has arisen ("Dispute Notice"), and within 30 (thirty) days from the date of the Dispute Notice, the Disputing Parties have failed to reach an amicable settlement on any Dispute, then such Dispute shall be referred to and finally resolved by arbitration in accordance with this Clause 23 (Governing Law and Jurisdiction).
- 23.3 The arbitral tribunal shall consist of 3 (three) arbitrators. The Party raising the Dispute (the claimant (or claimants jointly)) shall nominate 1 (one) arbitrator, and the Party(ies) against whom the Dispute has been raised (respondent (or respondents jointly)) shall nominate the other arbitrator. The 2 (two)

MUMBAI SOM A TOLLY & T

A CONTROL OF THE PARTY OF THE P

arbitrators so nominated by the Parties shall within 30 (thirty) days of the appointment of the second arbitrator agree upon a third arbitrator who shall act as the presiding arbitrator. If no agreement is reached within that period, the third arbitrator shall be appointed in accordance with the Rules to act as the presiding arbitrator. The decision of the arbitration panel shall be final and binding on the Parties.

- 23.4 The seat and venue of arbitration shall be Mumbai. This Agreement, including the arbitration agreement contained in this Clause 23 (*Governing Law and Jurisdiction*), shall be governed by and construed under the Laws of India and the courts in Mumbai shall have exclusive jurisdiction.
- 23.5 The language of arbitration shall be English.
- 23.6 For the purposes of the Rules, each Party consents to be joined to any arbitration commenced pursuant to this Agreement.
- 23.7 For the purposes of the Rules, each Party agrees to the consolidation of any two or more arbitrations commenced pursuant to this Agreement into a single arbitration.





DETAILS OF SUBSCRIBER

#	NAME AND DETAILS OF SUBSCRIBER	DETAILS OF DEMAT ACCOUNT	
		(DP ID & Client ID)	
1.	TATA AIG General Insurance Company Limited	DP ID : IN300167	
		Client ID :10007354	





SUBSCRIPTION SHARES DETAILS

#	NAME SUBSCRIBER	OF	NUMBER OF SHARES BEING SUBSCRIBED	PRICE PER SHARE (IN INR)	PRICE (IN INR)	PAN
1.	TATA AIG Ger Insurance Comp Limited	neral pany	14,18,400	423	59,99,83,200	AABCT3518Q





DEFINITIONS AND INTERPRETATION

1. <u>Definitions.</u> In this Agreement, the following words and expressions shall have the following meanings:

Act means the (Indian) Companies Act, 2013, as may from time to time be amended, re-enacted or replaced, read with all the rules and regulations made and notifications and circulars issued thereunder;

Board means the board of directors of the Company;

Business Day means a day other than a Saturday or Sunday or public holiday in Mumbai on which banks are not open in Mumbai for general commercial business;

CDSL shall mean Central Depository Services (India) Limited;

Closing means completion of the issuance and allotment of the Subscription Shares to the Subscriber, in accordance with Clause 5 (*Closing*) of this Agreement;

Closing Date has the meaning given in Clause 5.1 (Closing);

Company's Depository Participant shall mean Stellar Securities Private Limited, DP of Central Depository Services (I) Ltd., having DP ID 12085800 and registered office at 406, Sanjar Enclave, OPP PVR Milap Cinema, S V Road, Kandivli West, Mumbai, Maharashtra – 400 06;

Condition Precedent shall have the same meaning given to the term in Clause 4.1 (Condition Precedent);

CP Confirmation Certificate shall have the same meaning given to the term in Clause 4.5 (*Conditions Precedent*);

CP Satisfaction Certificate shall have the same meaning given to the term in Clause 4.4 (Conditions Precedent);

Depository shall have the meaning ascribed to the term under the Depositories Act, 1996 and, with respect to the Company, shall mean NSDL and CDSL;

Designated Bank Account shall have the same meaning given to the term in Clause 3.3 (Price);

Dispute shall have the meaning given to the term in Clause 23.2 (Governing Law and Jurisdiction);

Dispute Notice shall have the meaning given to the term in Clause 23.2 (Governing Law and Jurisdiction);

Disputing Parties shall have the meaning given to the term in Clause 23.2 (Governing Law and Jurisdiction);

Encumbrances means (a) any mortgage, charge (whether fixed or floating), pledge, lien, hypothecation, assignment, deed of trust, title retention, non-disposal undertakings, security interest or other encumbrance of any kind securing, or conferring any priority of payment in respect of, any obligation of any Person, including any right granted by a transaction which, in legal terms, is not the granting of security but which has an economic or financial effect similar to the granting of security under applicable Law, (b) purchase or option agreement or arrangement, (c) any adverse claim as to title, possession or use, and (d) any proxy, power of attorney, voting trust agreement, interest, option, right of first offer, refusal or transfer restriction in favour of any Person;

A STECHNOLOGIES IN THE PROPERTY OF THE PROPERT

MUMBAI * OIL * OIL *

Equity Share means the equity shares of the Company, having a par or nominal value of Rs. 2 each;

Form PAS-4 shall have the same meaning given to the term in Clause 2.1 (Subscription Shares);

Indemnifying Party shall have the meaning given to the term in Clause 10.1 (Indemnification);

Indemnified Person(s) shall have the meaning given to the term in Clause 10.1 (Indemnification);

IPO shall have the same meaning given to the term in Recital A;

Law means all applicable laws, by-laws, rules, regulations, orders, ordinances, protocols, codes, guidelines, policies, notices, directions, judgments, decrees, circulars, notifications, clarifications or other requirements or official directive or regulatory restriction or condition, or any similar form of decision of, or determination by, or interpretation of, having the force of law of any governmental authority or Person acting under the authority of any governmental authority or of any statutory or regulatory authority in India, whether in effect on the date of this Agreement or on the Closing Date in any jurisdiction and includes any practice or custom under any applicable Law and in each case as amended from time to time;

Long Stop Date means 15 (fifteen)] days from Execution Date, or such extended date which the Parties may mutually agree to in writing;

NSDL shall mean National Securities Depository Limited;

Price means 59,99,83,200 (Rupees Fifty Nine Crores Ninety Nine Lakhs Eighty Three Thousand Two Hundred only), being the aggregate price payable by the Subscriber for issuance and allotment of all the Subscription Shares calculated on the basis of ₹ 423 (Rupees Four hundred and twenty three only) per Subscription Share, as more particularly set out in **Schedule 2** (Subscription Shares Details) of this Agreement;

Proposed Transaction means the transaction contemplated by this Agreement;

Rules shall have the meaning given to the term in Clause 23.2 (Governing Law and Jurisdiction);

SEBI shall have the meaning given to the term in Recital A;

Subscription Shares shall mean in aggregate 14,18,400 Equity Shares of the Company which are proposed to be allotted to the Subscriber in accordance with the terms of this Agreement, as more particularly indicated in **Schedule 2** (Subscription Shares Details);

Subscriber's Warranties means the warranties given by the Subscriber pursuant to Clause 9 (Subscriber Warranties); and

Working Hours means 9:00 am to 6:00 pm in the relevant location on a Business Day.

- 2. <u>Interpretation.</u> In this Agreement, unless the context otherwise requires:
 - (a) references to a *Person* include any individual, firm, body corporate (wherever incorporated), government, state or agency of a state or any joint venture, association, partnership, works council or employee representative body (whether or not having separate legal personality);
 - (b) The recitals contained herein shall be deemed to be an integral part of this Agreement;
 - (c) When any day referenced in this Agreement is not a Business Day and a period referenced in this Agreement does not end on a Business Day, the reference to that day or to the last day of

MUMBAI MUMBAI MEMORE COMPANDED IN THE CO

NSURA

17

that period will be construed as a reference to the immediately preceding Business Day;

- (d) The rights and obligations of the Subscriber under this Agreement shall be exercised jointly;
- (e) A reference to a certified copy of a document means a copy certified to be a true, complete and up-to-date copy of the original document, in writing and signed by a director or the secretary of the Party delivering the document;
- (f) headings do not affect the interpretation of this Agreement; the singular shall include the plural and vice versa; and references to one gender include all genders;
- (g) references to any Indian legal term or concept shall, in respect of any jurisdiction other than India, be construed as references to the term or concept which most nearly corresponds to it in that jurisdiction;
- (h) any phrase introduced by the terms *including*, *include*, *in particular* or any similar expression shall be construed as illustrative and shall not limit the sense of the words preceding those terms; and
- (i) Reference to any legislation or Law or to any provision thereof shall include references to any such Law as it may, after the Effective Date, from time to time, be amended, supplemented or re-enacted, and any reference to statutory provision shall include any subordinate legislation made from time to time under that provision.
- 3. <u>Schedules and Exhibits.</u> The Schedules comprise schedules to this Agreement and form part of this Agreement.

NSUR

PART A

THE SHAREHOLDING PATTERN OF THE COMPANY AS AT THE EXECUTION DATE

#	Name of the Shareholder	No. of shares held	% of shareholding
1.	Pragnyat Pravin Lalwani	7,01,17,787	47.50
2.	Gautam Sampatraj Jain	7,01,17,787	47.50
3.	Florintree Nextech LLP	66,43,363	4.50
4.	Gopesh Vijayraj Modi	7,37,463	0.50
5.	Gautam Jain HUF	20	0.00
6.	Sunita Gautam Jain	30	0.00
7.	Pranati Ratnadeep Patil	20	0.00
8.	Ganesh Srinivasan	10	0.00
9.	Shweta Oswal	20	0.00
	TOTAL	14,76,16,500	100.00





PART B

THE SHAREHOLDING PATTERN OF THE COMPANY AS ON THE CLOSING DATE

#	Name of the Shareholder	No. of shares held	% of shareholding
1	Pragnyat Pravin Lalwani	7,01,17,787	46.60%
2	Gautam Sampatraj Jain	7,01,17,787	46.60%
3	Florintree Nextech LLP	66,43,363	4.42%
4	Gopesh Vijayraj Modi	7,37,463	0.49%
5	Gautam Jain HUF	20	0.00%
6	Sunita Gautam Jain	30	0.00%
7	Pranati Ratnadeep Patil	20	0.00%
8	Ganesh Srinivasan	10	0.00%
9	Shweta Oswal	20	0.00%
10	Tata AIG General Insurance Company Limited	14,18,400	0.94%
11	VQ FASTERCAP FUND II	7,09,200	0.47%
12	Valuequest India G.F.I.T Fund	7,09,200	0.47%
	TOTAL	15,04,53,300	100.00%





CP SATISFACTION CERTIFICATE

Date: [•]

[On the letterhead of the issuing Party]

To,

[insert name and address of the relevant Party]

Dear Sir(s),

We write with reference to the Share Subscription Agreement dated [●], ("Agreement") executed amongst the Company and the [•].

Capitalised terms and expressions used in this letter but not defined shall have the same meaning as ascribed to such terms and expressions in the Agreement.

This certificate is being issued pursuant to Clause 4.4 (Conditions Precedent) of the Agreement.

We hereby certify that all the Conditions Precedent set out at Clause 4 (Conditions Precedent) of the Agreement have been [fulfilled by us / modified in writing, by the [Subscribers]) prior to the Closing Date. The following documents evidencing the fulfilment of the Conditions Precedent applicable to us have been enclosed along with this letter:

[•]

Yours faithfully,

Signed and delivered for and on behalf of [•]:

Name :

Designation : [•]

21

CP CONFIRMATION CERTIFICATE

Date: [•]

[On the letterhead of the issuing Party]

To,

[insert address details]

Dear [●],

Re: CP Confirmation Certificate

We write with reference to the Share Subscription Agreement dated [●], ("Agreement") executed among the Company and [•].

Capitalised terms and expressions used in this letter but not defined shall have the same meaning as ascribed to such terms and expressions in the Agreement.

We refer to the CP Satisfaction Certificate dated [•] issued by the Company to us in accordance with Clause 4.5 (*Conditions Precedent*) of the Agreement. We hereby confirm fulfilment of the Conditions Precedent set out in Clause 4 (*Conditions Precedent*) of the Agreement.

Yours faithfully,

Signed and delivered for and on behalf of [insert name of issuing Party]:

Name

Designation

[•]

IN WITNESS WHEREOF this Agreement has been duly executed by the Parties and is intended to be and is hereby delivered on the date first above written.

Signed and delivered for and on behalf of the Company duly represented through its authorised representative:

Name: Pragnyat Pravin Lalwani

Title : Chairman and Managing Director

IN WITNESS WHEREOF this Agreement has been duly executed by the Parties and is intended to be and is hereby delivered on the date first above written.

Signed and delivered for and on behalf of **Tata AIG General Insurance Co. Ltd** duly represented through its authorised representative:



Name: Vinay Rao

Title : Chief Investment Officer

प्राप्त शास्त्र वास्त्र वास्त्र

GOVERNMENT OF MAHABASHIRA स्य दिन खुल है। साम्राह्म प्रतिकारी स्थापन SECURED BANK & TREASURY RECEIPT (e-SBT

21637588627589



Bank/Branch: SBI / 11688-SME BACKBAY RECLAMATION BRANCH

Pmt Txn id:

133353

Stationery No

XXXXXXXXXX7589

Pmt DtTime: 08-08-2025@04:59:58 Print DtTime: 08-08-2025@05:05:35

District:

7101/MUMBAI

Office Name: IGR182-BOM1_MUMBAI CITY 1 SUB REGISTRAR

ChallanIdNo: 00211688080825127818 GRAS GRN: MH006859998202526S

GRN Date:

08-08-2025@04:59:59

StDuty Schm: 0030045501

StDuty Amt: Rs 1199967/-(Rs One One, Nine Nine, Nine Six Seven Only)

RgnFee Schm

RgnFee Amt:

Article:

5(h)(A)(iv)/Agreement creating right and having monetary value

Prop Mvblty: Not Applicable

Consideration Rs 599983200/-

Prop Descr:

9 LALWANI INDUSTRIAL ESTATE, 14 KATRAK ROAD, WADALA WEST, 400031

Duty Payer:

PAN-AABCS1741G, SESHAASAI TECHNOLOGIES LIMITED

Other Party:

PAN-AADCV1630K, VALUEQUEST INVESTMENT ADVISORS PRIVATE LIMITED

Bank official-1 Name & Signature

Bank official-2 Name & Signature





e-SBTR IS VALID UPTO SIX MONTHS FROM THE DATE OF PAYMENT.



महाराष्ट्र MAHARASHTRA

PROPERTY AND PROPERTY OF THE PROPERTY

• 2025 • AUG 2015

DW 662808

This Stamp paper forms an integral part of the Share Subscription Agreement dated August 8, 2025 entered between Seshaasai Technologies Limited (hereinafter referred to as the "Company") and VQ Fastercap Fund II and Valuequest India G.I.F.T. Fund (hereinafter each referred to collectively as "Subscribers" and individually as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns)









28 JUL 20 Series 8 THE OWNER OF THE OWNER OWNER OF THE OWNER OWNER

HELD.

1000

This Stamp paper forms an integral part of the Share Subscription Agreement dated August 8, 2025 entered between Seshaasai Technologies Limited (hereinafter referred to as the "Company") and VQ Fastercap Fund II and Valuequest India G.I.F.T. Fund (hereinafter each referred to collectively as "Subscribers" and individually as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns)











2025

DW 662804

This Stamp paper forms an integral part of the Share Subscription Agreement dated August 8, 2025 entered between Seshaasai Technologies Limited (hereinafter referred to as the "Company") and VQ Fastercap Fund II and Valuequest India G.I.F.T. Fund (hereinafter each referred to collectively as "Subscribers" and individually as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns)



1







This Stamp paper forms an integral part of the Share Subscription Agreement dated August 8, 2025 entered between Seshaasai Technologies Limited (hereinafter referred to as the "Company") and VQ Fastercap Fund II and Valuequest India G.I.F.T. Fund (hereinafter each referred to collectively as "Subscribers" and individually as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns)



STATE OF



SHARE SUBSCRIPTION AGREEMENT

THIS AGREEMENT is made on August 08, 2025 ("Execution Date"), by and between:

 ENTITIES DESCRIBED UNDER SCHEDULE 1, (hereinafter each referred to collectively as "Subscribers" and individually as "Subscriber", which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns) of the FIRST PART;

AND

2. Seshaasai Technologies Limited, a company incorporated under the Laws of India under the provisions of the Companies Act, 1956, having company identification number U21017MH1993PLC074023 and having its registered office at 9, Lalwani Industrial Estate 14, Katrak Road, Wadala (West), Mumbai, Maharashtra, India, 400031, hereinafter referred to as the "Company" (which expression shall, unless it be repugnant to the context or meaning thereof, mean and include its successors and permitted assigns) of the SECOND PART.

The Subscribers and the Company shall hereinafter, collectively, be referred to as "Parties" and individually, as a "Party".

WHEREAS:

- A. The Company is a technology driven multi-location solutions provider offering payments solutions, communications and fulfilment solutions and IoT solutions and is proposing to launch an initial public offer ("IPO") of its securities. The Company has filed a draft red herring prospectus with the Securities Exchange Board of India ("SEBI") in relation to such IPO on December 27, 2024 ("DRHP").
- B. As of the Execution Date, the authorized share capital of the Company is ₹ 1,62,50,00,000 divided into 16,25,00,000 Equity Shares of ₹ 10 each and the paid-up share capital of the Company is ₹ 1,47,61,65,000 divided into 14,76,16,500 Equity Shares of ₹ 10 each. The shareholding pattern of the Company as at the Execution Date is set out in Part A of Schedule 4 (Shareholding Pattern of the Company As At The Execution Date).
- C. The Company, as a part of its pre-IPO placement, is offering its Equity Shares for subscription by the Subscribers and the Subscribers are desirous of making a financial investment in the Company by subscribing to the Subscription Shares for the Price, as set out against their respective names in **Schedule 2** (Subscription Shares Details).
- D. The Subscription Shares are proposed to be listed on the stock exchange(s) as part of IPO of the Company.
- E. The Parties are now entering into this Agreement to define their mutual rights and obligations in relation to the issue and allotment of the Subscription Shares to the Subscribers by the Company and other matters in connection therewith.

NOW, THEREFORE, THE PARTIES AGREE AS FOLLOWS:

1. DEFINITION AND INTERPRETATION

Unless a contrary intention appears and/or the context otherwise requires, in addition to the terms

SCHOOL STATE OF THE STATE OF TH

A THE SECOND

MUMBAI RS PY

defined elsewhere in this Agreement, the definitions listed in **Schedule 3** (*Definitions and Interpretations*) shall apply throughout this Agreement. The interpretation and/or construction of this Agreement shall be in accordance with the rules of interpretation set out in **Schedule 3** (*Definitions and Interpretation*).

2. SUBSCRIPTION SHARES

- 2.1 The Subscribers have agreed to subscribe to the Subscription Shares, as set out in **Schedule 2** (Subscription Shares Details) in accordance with this Agreement, basis the representations and warranties provided by the Company, and subject to receipt of the offer letter in Form PAS-4 in the format as set out in the Companies (Prospectus and Allotment of Securities) Rules, 2014 ("Form PAS-4"), from the Company and completion of the other actions as specified in Clause 4 (Conditions Precedent) below.
- 2.2 Each Subscriber's obligations under this Agreement are several, separate, and independent from those of any other Subscriber. No Subscriber shall be liable for the obligations, commitments, or liabilities of any other Subscriber, and the failure of any Subscriber to perform its obligations shall not relieve any other Subscriber of its own obligations under this Agreement.

3. PRICE

- 3.1 The Subscribers have agreed to pay the Price, in the proportion as set out in Schedule 2 (Subscription Shares Details), to subscribe to the Subscription Shares, for a total aggregate subscription amount of ₹ 59,99,83,200 (Rupees Fifty Nine Crores Ninety Nine Lakhs Eighty Three Thousand Two Hundred only). The resultant shareholding of the Subscribers in the Company, post completion of investment of the Price into the Company by the Subscribers shall be as set out in Part-B of Schedule 4 (Shareholding Pattern of the Company as on the closing date).
- 3.2 The issuance of the Subscription Shares shall be made by the Company on a private placement basis, in accordance with applicable Law.
- 3.3 The Subscribers shall pay, by electronic funds transfer, the Price on the Closing Date to the following bank account of the Company ("Designated Bank Account"), for the issuance and allotment of the Subscription Shares to the Subscribers on the Closing Date:

Account No:

000405164258

Name:

Seshaasai Technologies Limited

Customer ID:

602657624

IFSC code:

ICIC0000004

Branch:

ICICI Bank Limited, 215, Free Press House,

Free Press Marg, Nariman Point, Mumbai, 400021

Bank Name:

ICICI Bank Limited

3.4 Immediately upon receipt of the Price, the Company shall submit to the Subscribers a written confirmation of receipt showing the date and amount of the Price received.



FCHNOZOGOSTA WECHNOZOGOSTA WECHNOZ

B.

4. CONDITIONS PRECEDENT

- 4.1 The obligation of the Subscribers to subscribe to the Subscription Shares shall be conditional upon each of the following conditions (each a "Condition Precedent") having been fulfilled by the Company, to the reasonable satisfaction of the Subscribers, before the Closing Date:
 - (a) the Company having passed necessary resolutions of the Board and shareholders (*via* a special resolution), as specified under the Act and the Articles of Association of the Company for: (i) approving the issue and allotment of the Subscription Shares to the Subscribers for the Price as contemplated in this Agreement; and (ii) approving the draft letter of offer in Form PAS 4;
 - (b) delivery of certified true copies of the aforesaid resolutions of the Board and shareholders of the Company to the Subscribers;
 - (c) the Company having issued an offer or invitation to the Subscribers to subscribe to the Subscription Shares through issue of a private placement offer letter in the format as set out in Form PAS-4;
 - (d) The Company having filed, prior to the issuance of the offer letter in Form PAS-4, Form MGT-14 as prescribed under the Companies (Management and Administration) Rules, 2014 with the jurisdictional registrar of companies in respect of approving issuance of Equity Shares to the Subscribers; and
 - together with Form PAS-4, Company to deliver to the Subscribers, a valuation certificate issued by Registered Valuer, certifying the fair value of the Subscription Shares determined and issued in accordance with, and as required under (i) the Act (and rules thereunder), (ii) the Foreign Exchange Management Act, 1999 (and rules thereunder), and (iii) Section 56(2)(x) and Section 50CA of Income Tax Act read with Rule 11UA and Rule 11UAA of the Income Tax Rules, 1962; in each case, on which the Subscribers can rely.
- 4.2 If any of the Conditions Precedent are not satisfied, the Subscribers may, in writing, amend or extend the timeline for completion of any of the Conditions Precedent, by notice in writing to the Company, except to the extent of any mandatory requirements under applicable Law.
- 4.3 If at any time the Company becomes aware of any circumstances that will or are likely to give rise to the non-fulfilment of the Conditions Precedent by the Closing Date the Company shall inform the Subscribers in writing. In such case, the Parties shall co-operate fully with a view to procuring alternate arrangements to give effect to the commercial understanding of the Parties in relation to the transactions contemplated under this Agreement.
- Upon completion of the Conditions Precedent as aforesaid, the Company shall issue a certificate in the form annexed as **Schedule 5** ("**CP Satisfaction Certificate**") enclosing, where applicable, documentary evidence including certified true copies of all necessary documents evidencing fulfilment of each of the Conditions Precedent.
- 4.5 Upon fulfilment of all the Conditions Precedent, the Parties shall immediately proceed to Closing.

5. CLOSING

5.1 Subject to fulfilment of the Conditions Precedent, the Closing shall take place at the registered office



MUMBAI OF PART A COLOR

of the Company, or such other place as may be mutually agreed between the Parties. The "Closing Date" shall be the date on which Closing occurs, which date shall be: (a) a date that is not later than 3 (three) Business Days from the Execution Date; or (b) such other date as may be mutually agreed, in writing, between the Parties, but not later than the Long Stop Date.

- 5.2 On the Closing Date, the events set out in the following provisions of this Clause 5 (*Closing*) shall take place in the sequence set out below, each one conditional upon the other and not to be treated as mutually exclusive. Further, all transactions contemplated by this Agreement to be consummated at Closing shall be deemed to have occurred simultaneously and no such action shall be deemed to be consummated unless all such actions have been consummated. Closing shall be deemed to have occurred when each of the activities listed herein below have been completed and the Closing Date shall be deemed to be the date on which the last of the actions below are completed:
 - (a) The Subscribers shall remit the Price into the Designated Bank Account of the Company, by way of a wire transfer through normal banking channels;
 - (b) On the Closing Date, and simultaneously upon receipt of the Price, the Board or the Committee of the Board shall pass necessary resolutions, to the reasonable satisfaction of the Subscribers: (i) approving the issue and allotment of the Subscription Shares to the Subscribers, in dematerialized form, free of all Encumbrances; (ii) authorize issuance of instructions to the Company's depository participant to credit the Subscription Shares to the dematerialized account of the Subscribers (details of which are contained in **Schedule 1** (*Details of Subscribers*) below); and (iii) authorize the preparation and submission of the relevant filings related to the issuance and allotment of the Subscription Shares. The Company shall hand over to the Subscribers a certified true copy of the aforesaid resolution of the Board or the Committee of the Board.
 - (c) The Company shall make the necessary applications and filings with the Company's Depository Participant, providing irrevocable instructions for the issue of Subscription Shares to the Subscribers. The Company shall provide a copy of these instructions to the Subscribers.
- Notwithstanding anything to the contrary in this Agreement, if Closing does not occur as contemplated after any Subscriber remits its Subscription Amount, the Company shall, at such Subscriber's option and upon notice, refund the same within 5 (five) Business Days, and this Agreement shall stand terminated vis-à-vis such Subscriber with no further obligations or claims against it.
- 5.4 Subject to Clause 5.3 above, if the Closing has not occurred on or prior to the Long Stop Date, the Agreement shall terminate automatically (without any further action on the part of any of the Parties), unless Parties mutually agree in writing to extend the Long Stop Date in accordance with the terms hereof.
- 5.5 It is hereby clarified and agreed between the Parties that the failure of any Subscriber to pay its respective portion of the Price on or prior to the Closing Date shall not affect the rights and ability of the other Subscribers to invest their respective portions of the Price and the Company's obligation to issue and allot the Subscription Shares to such other Subscribers.

6. POST CLOSING OBLIGATION

6.1 The Company will take all reasonable efforts to ensure that the Subscription Shares are allotted by the



MUMBAI RESTMENT ADJUGORS ON THE THE PART ADJUGORS ON THE PART ADJUGORS O

Company's Depository Participant to the Subscribers no later than 5 (five) Business Days from the Closing Date. On allotment of Subscription Shares, the Company will provide an updated statement of beneficiary position to the Subscribers.

- After Closing, the Company shall complete all post-issue filings and compliances related to the issuance of the Subscription Shares within the time statutorily prescribed for the same, including (a) filing the return of allotment with the registrar of companies in Form PAS-3; and (b) to the extent applicable, filing the Single Master Form-FC-GPR or such other forms as may be prescribed in relation to issue and allotment of the Subscription Shares with the Reserve Bank of India in relation to the relevant Subscribers.
- 6.3 The Company agrees that the funds disbursed by the Subscribers for the subscription of the Subscription Shares shall not be utilized by the Company unless the allotment of the Subscription Shares have been completed and the return of allotment is filed with the jurisdictional registrar of companies, in accordance with the applicable Law.
- Within 30 (thirty) days of the financial statements of the Company for the quarter ending immediately post-Closing being disclosed / released by the Company as per applicable Laws, the Company shall provide to the Subscribers a valuation report (on a reliance basis) issued by an independent chartered accountant (in a form and substance substantially similar to the report being issued prior to Closing), which provides the fair market value of equity shares of the Company in accordance with Section 56(2)(x) and Section 50CA of Income Tax Act read with Rule 11UA and Rule 11UAA of the Income Tax Rules, 1962 as of the date of such quarter ending.
- 6.5 The Company shall provide the Subscribers or any of their authorised representatives and advisers such assistance, documentation and information and undertake all actions as may be reasonably required in connection with the filings and disclosures required or agreed to be made to the registrar of companies, SEBI, stock exchanges and any other disclosures required to be made under applicable Law pertaining to the transactions contemplated by this Agreement.

7. NON-COMPLETION OF INITIAL PUBLIC OFFERING

- 7.1 In the event the Subscription Shares post Closing are not listed on the NSE or the BSE within 60 (sixty) Business Days (or such longer period as may be mutually agreed between the Parties) from the Closing Date ("Listing Period"), if requested by the Subscribers (under this Agreement) within 30 (thirty) days of the Listing Period, and subject to Applicable Law, the Company shall, at its their discretion, either:
 - (a) purchase back the Subscribed Shares ("Buyback") and take all necessary steps, subject to Applicable Law to undertake the Buyback at the Price paid by the Subscribers in relation to the Subscribed Shares; or
 - (b) facilitate a purchase of the Subscribed Shares by a third party at the Price paid by the Subscriber in relation to the Subscribed Shares.

8. COMPANY WARRANTIES

8.1 The Company represents and warrants to each of the Subscribers that, as at the date of this Agreement and on the Closing Date:



MUMBAI SORS LIMING MUMBAI SORS CONTRACTOR OF STATE OF STA

- (a) it is validly incorporated, in existence and duly registered under the Laws of its jurisdiction and has full power and authority to conduct its business as conducted;
- (b) the Company is authorised by its articles of associations and has obtained all corporate authorisations and has all governmental, statutory, regulatory or other consents, licences and authorisations required to empower it to execute, deliver and perform its obligations under this Agreement and execution, delivery and performance of this Agreement constitutes legal, valid and binding obligation of the Company, enforceable against itself in accordance with its terms;
- (c) execution, delivery and performance of this Agreement will not (i) conflict with or violate any provision of its memorandum and articles of association, by-laws or equivalent constitutional documents, (ii) conflict with or contravene the provisions of or constitute a default under any documents (including IPO related documents and agreement executed by the Company), contracts, agreements, arrangements or any other instrument to which the Company is a party or which is applicable to it; (iii) result in a breach of any applicable Laws or regulations or of any order, decree or judgment of any court or any governmental or regulatory authority, applicable to the Company, or by which its assets and properties are bound or affected; or (iv) require the Company to obtain any authorisations, consents, waivers, registrations, declarations or filings with, or notice to any relevant authority or any person, except as expressly set forth in this Agreement or unless otherwise already obtained by the Company, copies of which have been delivered to the Subscribers;
- (d) issuance and allotment of the Subscription Shares by the Company to the Subscribers shall be duly and validly authorized by all necessary corporate actions of the Company and when issued, allotted and delivered in accordance with the terms of this Agreement, the Subscription Shares shall be duly and validly issued, fully paid and free and clear of any Encumbrance
- (e) there is no litigation, pending or threatened in writing, against the Company, challenging the validity or propriety of this Agreement or the transactions contemplated hereby or otherwise preventing the Company from entering into this Agreement and performing its obligations hereunder:
- (f) it has the requisite power and authority to enter into and perform this Agreement;
- (g) the Subscription Shares are duly authorised, validly issued and fully paid; and the Subscription Shares constitute 0. 49% of the shareholding of the Company immediately upon Closing. The shareholding percentage specified in **Part A** (Shareholding Pattern of the Company As At The Execution Date) and **Part B** (Shareholding Pattern As On The Closing Date) of **Schedule 4** provides true, correct and complete representations of the shareholding of the Company, constituting 100% (one hundred per cent) of the issued, subscribed, allotted and paid-up share capital of the Company, on a fully diluted basis, as of the Execution Date and the Closing Date, respectively;
- (h) the Subscription Shares will be issued in dematerialized form, free of all Encumbrances;
- (i) the Board has authorised the IPO of the Equity Shares of the Company and that the Company has filed the DRHP in relation to such IPO on December 27, 2024;



MUMBAI SON WILLIAM * OLI IN

- (j) the Company has not provided any unpublished price sensitive information to the Subscribers in contravention to the provisions of appliable Laws.
- The Company agrees and undertakes to notify the Subscribers, in writing and promptly, if it becomes aware of any fact, matter or circumstance prior to Closing which would cause any of the representation and warranties given by the Company under this Agreement, to become untrue, inaccurate or misleading in any material respect.

9. SUBSCRIBERS WARRANTIES

- 9.1 Each Subscriber warrants to the Company that as at the date of this Agreement and on the Closing
 - (a) it is validly incorporated, in existence and duly registered under the Laws of its jurisdiction and has full power and authority to conduct its business as conducted;
 - the execution, delivery and performance of this Agreement has been duly authorised and does not require any further corporate approvals and/ or authorizations and it has all necessary approvals required to be obtained by it for the execution, delivery and performance of this Agreement and has been validly executed by a duly authorised representative of the Subscriber. This Agreement constitutes a legal, valid and binding obligation of the Subscriber enforceable against it, in accordance with its terms;
 - (c) it has the requisite power and authority to enter into and perform this Agreement; and
 - (d) execution, delivery and performance of this Agreement will not (i) conflict with or violate any provision of its memorandum and articles of association, by-laws or equivalent constitutional documents or (ii) conflict with or contravene the provisions of or constitute a default under any documents, contracts, agreements, arrangements or any other instrument to which the Subscriber is a party or which is applicable to it; (iii) result in a breach of any Laws or regulations or of any registration, license, permit, consent, order, decree or judgment of any court or any governmental or regulatory authority, applicable to the Subscriber, or by which its assets and properties are bound or affected; or (iv) require the Subscriber to obtain any authorisations, consents, waivers, registrations, declarations or filings with, or notice to any relevant authority or any person, except as expressly set forth in this Agreement or unless otherwise already obtained by the Subscriber, copies of which have been delivered to the Company; and
 - (e) there is no litigation, pending or threatened in writing, against the Subscriber, challenging the validity or propriety of this Agreement or the transactions contemplated hereby or otherwise preventing the Subscriber from entering into this Agreement and performing its obligations hereunder.

10. INDEMNIFICATION

10.1 The Company ("Indemnifying Party") shall indemnify, defend and hold harmless the Subscribers, their respective investment manager, trustees, directors, officers, representatives, employees, successors and agents as applicable, ("Indemnified Persons") against any and all direct and actual

SSICH STMENT ADVISORS PRIVATED AT CHANCH STATED

MUMBAI DO MUMBAI

losses, damages, charges, and reasonable costs and expenses, fines, interest and penalties, imposed on, sustained, incurred or suffered by the Indemnified Persons on or after the Closing Date, directly arising out of, or directly in connection with:

- (a) any breach of, or misrepresentation in, any of the representations and warranties of the Company as set out in Clause 8.1 (Company Warranties);
- (b) any breach of covenants and undertaking provided by the Indemnifying Party under this Agreement; and/or
- (c) wilful misconduct, fraud, or gross negligence by the Indemnifying Party in connection with the transactions contemplated under this Agreement; and/or

Each, an "Indemnity Event".

Provided that if any claim (or part thereof) arises on account of a breach of any warranty or representation made by the Indemnified Parties, the Indemnifying Party shall not be liable to indemnify the Indemnified Parties in respect of such claim (or part thereof).

- To the extent the payment by the Indemnifying Party of any indemnification payment pursuant to the provisions of Clause 10 (*Indemnification*) is subject to receipt of approvals by the Indemnifying Party from governmental authorities, the Indemnifying Party shall apply to such governmental authorities to procure all such approvals and shall make all applications and take all steps required to obtain the same and the Indemnified Parties shall provide all necessary documents, support and co-operation as requested by the Indemnifying Party.
- 10.3 Notwithstanding anything contained in this Agreement, the Indemnifying Party shall not be liable to indemnify an Indemnified Party for any amount in excess of the portion of the Price paid by the respective Indemnified Party, save and except if the Indemnity Event is as set out in Clause 10.1 (c).
- Notwithstanding anything contained in this Agreement, each Indemnified Person expressly agrees and acknowledges that the indemnification rights contained in this Clause 10 are conditional upon the said indemnification rights being the sole monetary remedy for each Indemnified Person and accordingly, each Indemnified Person hereby expressly and specifically waives any right that it or any of them may have under Law, contract or equity to any other form of monetary remedy. Subject to occurrence of Closing, the indemnity under Clause 10.1(a) and10.1(c) shall survive for a period of 10 years from Closing Date.
- 10.5 Upon the occurrence of an indemnifiable event as set out in Clause 10.1 (*Indemnification*), the amount payable shall be grossed up to reflect the percentage of shares held by such Subscriber on a fully diluted basis at the time of such claim by the Subscriber.
- 10.6 General procedure for claims. If any Indemnified Person seeks indemnification under this Agreement, it shall, within a period of 30 (thirty) days from the occurrence of the Indemnity Event, notify in writing (the "Claim Notice") the Indemnifying Person of the Losses for which the Indemnified Person is asserting an indemnification claim ("Indemnification Claim Amount"). The Parties agree that a delay in providing the Claim Notice under this Clause 10.6 will not relieve the Indemnifying Person of its indemnification obligation, save and except to the extent of any increase in Loss caused due to the failure of the Indemnified Person to provide the Claim Notice as per the timelines prescribed in



this Clause 10.6.

10.7 A claim set forth in a Claim Notice will be conclusively deemed to be the liability of the Indemnifying Person (and payable by the Indemnifying Person) if: (a) the Indemnifying Person does not dispute the Claim Notice within a period of 10 (ten) days from the date of issue of the Claim Notice; or (b) the liability of the Indemnifying Person in respect of such claim is resolved by mutual agreement of the Indemnifying Person and the Indemnified Person. The Indemnifying Person shall pay the Indemnification Claim Amount to the Indemnified Person within 30 (thirty) days after such claim becomes payable in accordance with this Clause 10.7.

10.8 Procedure for Third Party Claims.

- (a) After receipt by the Indemnified Person of notice of the commencement of any action, suit, arbitration or other proceeding in relation to a Loss or an Indemnity Event involving a third party including a proceeding before a Governmental Authority in connection with an Indemnity Event (a "Third Party Claim"), such Indemnified Person shall as soon as commercially practicable, but in any event within 15 (fifteen) days of receipt of such Third Party Claim, give written notice of such Third Party Claim ("Third Party Claim Notice") to the Indemnifying Person providing all details, to the extent available, regarding the circumstances then known to the Indemnified Person. The Parties agree that a delay in providing the Third Party Claim Notice under this Clause 10.8(a) will not relieve the Indemnifying Person of its indemnification obligation in respect of a Third Party Claim, save and except to the extent of any increase in Loss caused due to the failure of the Indemnified Person to provide the Third Party Claim Notice as per the timelines prescribed in this Clause 10.8(a). Provided that in the event such delay by the Indemnified Party has been the sole cause of the complete failure of the defense of the Third Party Claim, then the Loss arising out of such Third-Party Claim shall not be required to be indemnified by the Indemnifying Person.
- (b) The Indemnifying Person shall, subject to Clause 10.8(d), have the right, exercisable by issuing a written notice to the Indemnified Person promptly, and in any event within 15 (fifteen) Business Days from the date of receipt of the Third Party Claim Notice, to assume the sole defence of any Third Party Claim at such Indemnifying Person's expense, provided that the Indemnifying Person shall: (i) consult, in good faith, with the Indemnified Person in relation to the conduct of the Third Party Claim; (ii) keep the Indemnified Person appropriately informed of all matters pertaining to such action
- (c) Where the Indemnifying Person elects to assume the defence of the Third-Party Claim, the Indemnified Person shall provide reasonable cooperation, at the Indemnifying Person's expense, with the Indemnifying Person and its legal advisors/ representatives. Where the Indemnifying Person is assuming defence of the Third Party Claim, the Indemnified Persons shall exercise reasonable efforts to provide all the relevant information and documents, and the representatives of the Indemnified Persons be reasonably available, for the Indemnifying Persons, its counsel and advisors for conducting the defence of the Third Party Claim, within a reasonable period of time, bearing in mind the overall timelines and strategy with respect to the defence of the Third Party Claim. The Indemnifying Person will not enter into any settlement, compromise, order or judgement in respect of any such Third Party Claim without the prior written consent of the Indemnified Person, which consent shall not be unreasonably withheld.



MUMBAJ OGESTMENT A COLT LINES TMENT A COLT LINES TM

- (d) Indemnifying Person will not have the right under Clause 10.8(b) to assume the sole defence of any Third Party Claim where: (i) the Third Party Claim seeks an injunction or equitable relief against any of the Indemnified Persons which restricts them from conducting their activities in the ordinary course of business; or (ii) the Third Party Claim involves a criminal liability on any of the Indemnified Persons; or (iii) the Indemnifying Person is not entitled to take up the defense under Applicable Law.
- (e) If the Indemnifying Person elects not to assume defence of the Third Party Claim, or the Indemnifying Person is not entitled to assume defence of the Third Party Claim as per Clause 10.8(d), the Indemnified Person will have the right to defend such Third Party Claim at the expense of the Indemnifying Person, and the Indemnifying Person shall cooperate, at its own expense, with the Indemnified Person and its legal advisors, in relation thereto. The Indemnified Person shall keep the Indemnifying Person appropriately informed of all matters pertaining to such Third Party Claim and shall conduct such Third Party Claim in a reasonable manner. The Indemnifying Person shall be deemed to have consented to and be bound by the manner in which such defence is assumed by Indemnified Person and the same will be binding upon the Indemnifying Person, and the Indemnified Person may enter into any judgment, compromise or settlement without the prior written permission of the Indemnifying Person (without prejudice to the indemnification rights of the Indemnified Person under this Agreement).
- (f) The Parties agree to cooperate with each other in connection with the status, defence, negotiation and/or settlement of any such Third Party Claim. Where the Indemnified Person is assuming sole defence of the Third Party Claim, the Indemnifying Persons shall ensure that all the relevant information and documents be provided, and the representatives of the Company/ Seller (as applicable) be reasonably available, for the Indemnified Persons, its counsel and advisors for conducting the defence of the Third Party Claim, within a reasonable period of time, bearing in mind the overall timelines and strategy with respect to the defence of the Third Party Claim.
- (g) It is hereby clarified that the Indemnifying Persons' obligation to indemnify the Indemnified Person in relation to a Third Party Claim pursuant to this Agreement will arise within 30 (thirty) days of the Indemnified Person incurring any actual expense in relation to such Third Party Claim (including costs and expenses for defending any such Third Party Claim (which defence will be undertaken as per this Agreement) and/or payments made pursuant to an order (interim or final), judgment, settlement or compromise (subject to such settlement or compromise being in accordance with this Agreement)).
- 10.9 <u>Miscellaneous</u>. Notwithstanding anything contained in this Agreement:
 - (a) The Indemnifying Person shall be required to indemnify the Indemnified Person only for actual Losses incurred or suffered. However, the Indemnifying Persons will not be liable for any loss of profit or loss of opportunity incurred/suffered by the Indemnified Persons.
 - (b) The Indemnified Persons shall not be entitled to recover more than once in respect of the same Loss.
- 10.10 The Indemnifying Persons shall not be liable to indemnify the Indemnified Persons under this Agreement for a Loss if the failure or breach giving rise to such Loss is capable of remedy and the



THOUGE STATE OF THE PROPERTY O

same has been remedied within a period of 30 (thirty) days from the date of notice provided by the Indemnified Persons to the Indemnifying Persons in writing of any claim for indemnity.

11. CONFIDENTIALITY

The Parties hereto acknowledge that the terms of this Agreement, including its existence shall be considered confidential information and shall not be disclosed by the Parties hereto to any third party except (i) any disclosure in relation to the IPO, including disclosures to be made in any Offer Document filed with SEBI, ROC or any other governmental or regulatory authority; (ii) any disclosure of information required by applicable Law, or any governmental or regulatory authority not in relation to the IPO; (iii) to its affiliates and investors that are bound by appropriate confidentiality obligations; (iv) the disclosure of information, which, at the date of disclosure, is in the public domain; (v) any disclosure of information to a Party's professional advisors being subject to the confidentiality obligations contained in this Agreement; and (vi) information which is independently developed by the recipient from information provided or obtained other than pursuant to or in connection with this Agreement.

12. ASSIGNMENT

- 12.1 Unless the Parties agree in writing, no person shall assign, transfer, charge or otherwise deal with all or any of its rights under this Agreement nor grant, declare, create or dispose of any right or interest in it, and any purported assignment in contravention of this Clause 12 (Assignment) shall be void.
- Notwithstanding anything contained in this Agreement, the provisions of this Clause 12 (*Assignment*) shall fall away on listing of equity shares of the Company.

13. TRANSFER

13.1 Subject to lock-in requirements under applicable Law, the Subscription Shares are freely transferable and are free from any restrictions under the articles of association of the Company or any other agreement to which the Company is a party.

14. FURTHER ASSURANCES

- 14.1 Each of the Parties shall, execute such further documents as may be required by Law or as may be necessary to implement and give effect to this Agreement.
- 14.2 Each of the Parties shall comply with all obligations under this Agreement which are expressed to apply to the respective Parties.

15. COSTS

- 15.1 The Parties shall each be responsible for their own costs, charges and other expenses incurred in connection with the Proposed Transaction. The Company shall be responsible to bear the stamp duty payable on this Agreement only to the extent that the document is executed and retained in Mumbai.
- 15.2 The stamp duty payable in relation to the issuance of the Subscription Shares (including on the allotment) shall be borne by the Company.

16. NOTICES



AT ADVIS

EGHNOZOGRASIA THOSE, OBJENESTM

- Any notice in connection with this Agreement shall be in writing in English and delivered by hand, e-mail, registered post or courier using an internationally recognised courier company. A notice shall be effective upon receipt and shall be deemed to have been received (i) at the time of delivery, if delivered by hand, registered post or courier or (ii) at the time of transmission if delivered by e-mail provided that in either case, where delivery occurs outside Working Hours, notice shall be deemed to have been received at the start of Working Hours on the next following Business Day. Notwithstanding the foregoing, the Parties hereby agree that every notice or communication must also be necessarily given by email.
- 16.2 The addresses and other details of the Parties for the purpose of Clause 16.1 (Notices) are:

a) If to Subscriber 1:

Name: VQ FasterCap Fund II

Address: C/o ValueQuest Investment Advisors Private Limited; Quest, 1073, Rajabhau Desai Marg, Behind Beau Monde Towers, Prabhadevi, Mumbai City, Maharashtra, India, 400025

Attention: Mr. Chirag Shah

Email : chirag.shah@valuequest.in

b) If to Subscriber 2:

Name: ValueQuest India G.I.F.T. Fund

Address: C/o ValueQuest Investment Advisors Private Limited (IFSC Branch); Unit No.B-107, Nila Spaces, Block 11, Road 1A, Plot T1-T4, GIFT SEZ, GIFT CITY, Gandhinagar, Gujarat-382050

Attention: Sshivani Sharma/Rahul Dua

Email: giftcompliance@valuequest.in with cc to sshivani@valuequest.in and rahul@valuequest.in

c) If to the Company:

Name: Seshaasai Technologies Limited (formerly Seshaasai Business Forms Limited)

Address: 9, Lalwani Industrial Estate 14, Katrak Road, Wadala (West), Mumbai - 400 031,

Maharashtra, India

Attention: Manali Siddharth Shah

Email: companysecretary@seshaasai.com

17. TERM AND TERMINATION

17.1 This Agreement shall come into full force and effect from the Execution Date and shall remain valid and binding on the Parties until such time that it is terminated in accordance with Clause 17.2 (*Term and Termination*).





- 17.2 This Agreement shall be terminated prior to Closing in the following manner:
 - (a) in accordance with Clause 5.3 (*Closing*) i.e., if the Closing has not occurred on or prior to the Long Stop Date; or
 - (b) By mutual written agreement of the Parties.
- 17.3 Notwithstanding anything provided in this clause or otherwise in this Agreement, the termination of this Agreement shall not relieve any Party of any obligations or liabilities accrued prior to the date of termination.
- 17.4 It is hereby agreed that in the event the Agreement is terminated prior to Closing, in terms of Clause 17.2 above, the Parties undertake to use their reasonable efforts to unwind all actions initiated or taken under the provisions of this Agreement in order to achieve Closing, as soon as possible.
- 17.5 Clause 1 (Definitions and Interpretation), Clause 10 (Confidentiality), Clause 15 (Costs), Clause 16 (Notices), Clause 17.3 (Term and Termination), Clause 17.4 (Term and Termination), Clause 17.5 (Term and Termination), Clause 18 (Whole Agreement and No Deed of Adherence), Clause 19 (Waivers, Rights and Remedies), Clause 20 (Counterparts), Clause 21 (Variations), Clause 22 (Invalidity), Clause 23 (Governing Law and Jurisdiction) shall survive termination of this Agreement;

18. WHOLE AGREEMENT AND NO DEED OF ADHERENCE

- 18.1 This Agreement sets out the whole agreement between the Parties in respect of the subscription Shares and supersede any prior agreement (whether oral or written) relating to the transaction contemplated under this Agreement.
- 18.2 The Company hereby confirms that based on the waiver letter received from the existing investors of the Company, the Subscribers need not execute a deed of adherence to the Share Subscription Agreement.

19. WAIVERS, RIGHTS AND REMEDIES

19.1 Except as expressly provided in this Agreement, no failure or delay by any party in exercising any right or remedy relating to this Agreement shall affect or operate as a waiver or variation of that right or remedy or preclude its exercise at any subsequent time. No single or partial exercise of any such right or remedy shall preclude any further exercise of it or the exercise of any other remedy.

20. COUNTERPARTS

20.1 This Agreement may be executed in any number of counterparts, and by each party on separate counterparts. Each counterpart is an original, but all counterparts shall together constitute one and the same instrument. The delivery of a signed counterparts of this Agreement by facsimile transmission or e-mail in "portable document format" (.pdf) shall be as effective as signing and delivering the counterparts in person.

21. VARIATIONS

21.1 No amendment of this Agreement shall be valid unless it is in writing and duly executed by or on behalf of all of the Parties to it.

CSYON GO TESC BRANCH WATER

THE WOLDS ESTABLISHED TO THE STATE OF THE ST

22. SEVERABILITY

22.1 Each of the provisions of this Agreement is severable. If any such provision is held to be or becomes invalid or unenforceable in any respect under the Law of any jurisdiction, it shall have no effect in that respect and the Parties shall use all reasonable efforts to replace it in that respect with a valid and enforceable substitute provision, the effect of which is as close to its intended effect as possible.

23. GOVERNING LAW AND JURISDICTION

- 23.1 This Agreement and any obligations arising out of or in connection with this Agreement shall be governed by and interpreted in accordance with Indian Law, without regard to conflict of law principles.
- Any dispute or differences between the Parties ("Disputing Parties") arising out of or in connection with this Agreement, including any question regarding its existence, validity or termination, and any dispute relating to any obligations arising out of or in connection with it (the "Dispute") shall be referred to and finally resolved by arbitration in accordance with the provisions of the Arbitration and Conciliation Act, 1996 or any statutory modification or re-enactment thereof for the time being in force ("Rules"). If a Disputing Party gives the other Disputing Party(ies) notice that a Dispute has arisen ("Dispute Notice"), and within 30 (thirty) days from the date of the Dispute Notice, the Disputing Parties have failed to reach an amicable settlement on any Dispute, then such Dispute shall be referred to and finally resolved by arbitration in accordance with this Clause 23 (Governing Law and Jurisdiction).
- The arbitral tribunal shall consist of 3 (three) arbitrators. The Party raising the Dispute (the claimant (or claimants jointly)) shall nominate 1 (one) arbitrator, and the Party(ies) against whom the Dispute has been raised (respondent (or respondents jointly)) shall nominate the other arbitrator. The 2 (two) arbitrators so nominated by the Parties shall within 30 (thirty) days of the appointment of the second arbitrator agree upon a third arbitrator who shall act as the presiding arbitrator. If no agreement is reached within that period, the third arbitrator shall be appointed in accordance with the Rules to act as the presiding arbitrator. The decision of the arbitration panel shall be final and binding on the Parties.
- 23.4 The seat and venue of arbitration shall be Mumbai. This Agreement, including the arbitration agreement contained in this Clause 23 (*Governing Law and Jurisdiction*), shall be governed by and construed under the Laws of India and the courts in Mumbai shall have exclusive jurisdiction.
- 23.5 The language of arbitration shall be English.
- 23.6 For the purposes of the Rules, each Party consents to be joined to any arbitration commenced pursuant to this Agreement.
- For the purposes of the Rules, each Party agrees to the consolidation of any two or more arbitrations commenced pursuant to this Agreement into a single arbitration.



THE THEN THE PARTY OF THE PARTY

DETAILS OF SUBSCRIBERS

#	NAME AND DETAILS OF SUBSCRIBERS	DETAILS OF DEMAT ACCOUNT
1.	VQ FASTERCAP FUND II, a scheme of Valuequest India	Name: VQ FASTERCAP FUND II
	Investment Trust, a Category III Alternative Investment	DP: ICICI BANK LIMITED (NSDL)
	Fund, having Reg. No. IN/AIF3/24-25/1508 and PAN	DP ID : IN301348
	AAETV4585H, and acting through its investment manager,	BO ID: 20739609
	ValueQuest Investment Advisors Private Limited (CIN:	
	U65991MH2010PTC198452), a company incorporated	
	under the Companies Act, 1956 and having its registered	
	office at Quest, 1073, Rajabhau Desai Marg, Behind Beau	
	Monde Towers, Prabhadevi, Mumbai City, Maharashtra,	
	India – 400025 ("Subscriber 1")	
2.	VALUEQUEST INDIA G.I.F.T. FUND, a category III	Name: VALUEQUEST INDIA
	alternative investment fund registered with the International	G.I.F.T. FUND
	Financial Services Centre Authority bearing registration no.	DP : ICICI BANK LIMITED (NSDL)
	IFSC/AIF3/2024-25/0187, through its fund management	DP ID: IN301348
	entity ValueQuest Investment Advisors Private Limited	
	(IFSC Branch) having its branch office at Unit No. B-107,	
	Nila Spaces, Plot No. T1-T4, Road 1A, Block 11, Zone 1,	
	SEZ-PA, Gandhinagar - 382050, Gujarat, India, that is	
	registered with IFSCA as a Registered (Non-Retail) Fund	
	Management Entity bearing registration no.	
	IFSCA/FME/II/2024-25/133, regulated by the IFSCA	
	(Fund Management) Regulations, 2025 ("Subscriber 2")	





SUBSCRIPTION SHARES DETAILS

#	NAME OF SUBSCRIBERS	NUMBER OF SHARES BEING SUBSCRIBED	PRICE PER SHARE (IN ₹)	PRICE (IN ₹)
1.	VQ FasterCap Fund II	7,09,200	423	29,99,91,600
2.	ValueQuest India G.I.F.T. Fund	7,09,200	423	29,99,91,600
	Total	14,18,400		59,99,83,200



ESTMENT AND THE STIMENT AND TH

DEFINITIONS AND INTERPRETATION

1. <u>Definitions.</u> In this Agreement, the following words and expressions shall have the following meanings:

Act means the (Indian) Companies Act, 2013, as may from time to time be amended, re-enacted or replaced, read with all the rules and regulations made and notifications and circulars issued thereunder;

Board means the board of directors of the Company;

Business Day means a day other than a Saturday or Sunday or public holiday in Mumbai on which banks are not open in Mumbai for general commercial business;

CDSL shall mean Central Depository Services Limited;

Closing means completion of the issuance and allotment of the Subscription Shares to the Subscribers, in accordance with Clause 5 (*Closing*) of this Agreement;

Closing Date has the meaning given in Clause 5.1 (Closing);

Company's Depository Participant shall mean Stellar Securities Private Limited, DP of Central Depository Services (I) Ltd., having DP ID 12085800 and registered office at 406, Sanjar Enclave, OPP PVR Milap Cinema, S V Road, Kandivli West, Mumbai, Maharashtra – 400 06;

Condition Precedent shall have the same meaning given to the term in Clause 4.1 (Condition Precedent):

CP Satisfaction Certificate shall have the same meaning given to the term in Clause 4.4 (Conditions Precedent);

Depository shall have the meaning ascribed to the term under the Depositories Regulation Act, 1996 and, with respect to the Company, shall mean NSDL and CDSL;

Designated Bank Account shall have the same meaning given to the term in Clause 3.3 (Price);

Dispute shall have the meaning given to the term in Clause 23.2 (Governing Law and Jurisdiction);

Dispute Notice shall have the meaning given to the term in Clause 23.2 (Governing Law and Jurisdiction);

Disputing Parties shall have the meaning given to the term in Clause 23.2 (Governing Law and Jurisdiction);

Encumbrances means (a) any mortgage, charge (whether fixed or floating), pledge, lien, hypothecation, assignment, deed of trust, title retention, non-disposal undertakings, security interest or other encumbrance of any kind securing, or conferring any priority of payment in respect of, any obligation of any Person, including any right granted by a transaction which, in legal terms, is not the





granting of security but which has an economic or financial effect similar to the granting of security under applicable Law, (b) purchase or option agreement or arrangement, (c) any adverse claim as to title, possession or use, and (d) any proxy, power of attorney, voting trust agreement, interest, option, right of first offer, refusal or transfer restriction in favour of any Person;

Equity Share means the equity shares of the Company, having a par or nominal value of ₹ 10 (Indian Rupees Ten) each;

Form PAS-4 shall have the same meaning given to the term in Clause 2.1 (Subscription Shares);

Indemnifying Party shall have the meaning given to the term in Clause 10.1(Indemnification);

Indemnified Person(s) shall have the meaning given to the term in Clause 10.1 (Indemnification);

IPO shall have the same meaning given to the term in Recital A;

Law means all applicable laws, by-laws, rules, regulations, orders, ordinances, protocols, codes, guidelines, policies, notices, directions, judgments, decrees, circulars, notifications, clarifications or other requirements or official directive or regulatory restriction or condition, or any similar form of decision of, or determination by, or interpretation of, having the force of law of any governmental authority or Person acting under the authority of any governmental authority or of any statutory or regulatory authority in India, whether in effect on the date of this Agreement or on the Closing Date in any jurisdiction and includes any practice or custom under any applicable Law and in each case as amended from time to time;

Long Stop Date means 15 (fifteen)] days from Execution Date, or such extended date which the Parties may mutually agree to in writing;

NSDL shall mean National Securities Depositories Limited;

Price means 59,99,83,200 (Rupees Fifty Nine Crores Ninety Nine Lakhs Eighty Three Thousand Two Hundred only), being the aggregate price payable collectively by the Subscribers for issuance and allotment of all the Subscription Shares calculated on the basis of ₹ 423 (Rupees Four hundred and twenty three only) per Subscription Share, as more particularly set out in **Schedule 2** (*Subscription Shares Details*) of this Agreement;

Proposed Transaction means the transaction contemplated by this Agreement;

Rules shall have the meaning given to the term in Clause 23.2 (Governing Law and Jurisdiction);

SEBI shall have the meaning given to the term in Recital A;

Subscription Shares shall mean in aggregate 14,18,400 Equity Shares of the Company which are proposed to be allotted to the Subscribers in accordance with the terms of this Agreement, as more particularly indicated in Schedule 2 (Subscription Shares Details);

Subscriber's Warranties means the warranties given by the Subscribers pursuant to Clause 9 (Subscribers Warranties); and

Working Hours means 9:00 am to 6:00 pm in the relevant location on a Business Day.

IFSC BRANCH WITH CHINE

WINDS RESTMENT TO INSORG

- 2. <u>Interpretation.</u> In this Agreement, unless the context otherwise requires:
 - (a) references to a *Person* include any individual, firm, body corporate (wherever incorporated), government, state or agency of a state or any joint venture, association, partnership, works council or employee representative body (whether or not having separate legal personality);
 - (b) The recitals contained herein shall be deemed to be an integral part of this Agreement;
 - (c) When any day referenced in this Agreement is not a Business Day and a period referenced in this Agreement does not end on a Business Day, the reference to that day or to the last day of that period will be construed as a reference to the immediately preceding Business Day;
 - (d) A reference to a certified copy of a document means a copy certified to be a true, complete and up-to-date copy of the original document, in writing and signed by a director or the secretary of the Party delivering the document;
 - (e) headings do not affect the interpretation of this Agreement; the singular shall include the plural and vice versa; and references to one gender include all genders;
 - (f) references to any Indian legal term or concept shall, in respect of any jurisdiction other than India, be construed as references to the term or concept which most nearly corresponds to it in that jurisdiction;
 - (g) any phrase introduced by the terms *including*, *include*, *in particular* or any similar expression shall be construed as illustrative and shall not limit the sense of the words preceding those terms; and
 - (h) Reference to any legislation or Law or to any provision thereof shall include references to any such Law as it may, after the Effective Date, from time to time, be amended, supplemented or re-enacted, and any reference to statutory provision shall include any subordinate legislation made from time to time under that provision.
 - 3. <u>Schedules and Exhibits.</u> The Schedules comprise schedules to this Agreement and form part of this Agreement.





PART A

THE SHAREHOLDING PATTERN OF THE COMPANY AS AT THE EXECUTION DATE

#	Name of the Shareholder	No. of shares held	% of shareholding
1.	Pragnyat Pravin Lalwani	7,01,17,787	47.50
2.	Gautam Sampatraj Jain	7,01,17,787	47.50
3.	Florintree Nextech LLP	66,43,363	4.50
4.	Gopesh Vijayraj Modi	7,37,463	0.50
5.	Gautam Jain HUF	20	0.00
6.	Sunita Gautam Jain	30	0.00
7.	Pranati Ratnadeep Patil	20	0.00
8.	Ganesh Srinivasan	10	0.00
9.	Shweta Oswal	20	0.00
	TOTAL	14,76,16,500	100.00



MUMBAI * OL'

PART B

THE SHAREHOLDING PATTERN OF THE COMPANY AS ON THE CLOSING DATE

#	Name of the Shareholder	No. of shares held	% of shareholding
1	Pragnyat Pravin Lalwani	7,01,17,787	46.60%
2	Gautam Sampatraj Jain	7,01,17,787	46.60%
3	Florintree Nextech LLP	66,43,363	4.42%
4	Tata AIG General Insurance Company Limited	14,18,400	0.94%
5	Gopesh Vijayraj Modi	7,37,463	0.49%
6	VQ Fastercap Fund II	7,09,200	0.47%
7	Valuequest India G.F.I.T Fund	7,09,200	0.47%
8	Gautam Jain HUF	20	0.00%
9	Sunita Gautam Jain	30	0.00%
10	Pranati Ratnadeep Patil	20	0.00%
11	Ganesh Srinivasan	10	0.00%
12	Shweta Oswal	20	0.00%
	TOTAL	15,04,53,300	100.00%



CP SATISFACTION CERTIFICATE

Date: [•]

[On the letterhead of the issuing Party]

To,

[insert name and address of the relevant Party]

Dear Sir(s),

We write with reference to the Share Subscription Agreement dated [●], ("Agreement") executed among the Company and the [•].

Capitalised terms and expressions used in this letter but not defined shall have the same meaning as ascribed to such terms and expressions in the Agreement.

This certificate is being issued pursuant to Clause 4.4 (Conditions Precedent) of the Agreement.

We hereby certify that all the Conditions Precedent set out at Clause 4 (Conditions Precedent) of the Agreement have been [fulfilled by us / modified in writing, by the [Subscribers]) prior to the Closing Date. The following documents evidencing the fulfilment of the Conditions Precedent applicable to us have been enclosed along with this letter:

[•]

Yours faithfully,

Signed and delivered for and anticalf of [•]:

Name

[•]

Designation

[•]

IN WITNESS WHEREOF this Agreement has been duly executed by the Parties and is intended to be and is hereby delivered on the date first above written.

Signed and delivered for and on behalf of the Company duly represented through its authorised representative:

Name:

Pragnyat Pravin Lalwani

Title

Chairman and Managing Director

IN WITNESS WHEREOF this Agreement has been duly executed by the Parties and is intended to be and is hereby delivered on the date first above written.

Signed and delivered on behalf of VQ FASTERCAP FUND II duly represented through its authorised representative:

Name: Ravindra Raichand Dharmshi

Title : Director

IN WITNESS WHEREOF this Agreement has been duly executed by the Parties and is intended to be and is hereby delivered on the date first above written.

Signed and delivered on behalf of Valuequest India G.F.I.T Fund duly represented through its authorised

MENT ADVIS

IFSC BRANC

representative:

Name:

Sshivani Panwar Sharma

Title

Principal Officer